Author's response to reviews

Title: MiR-9, miR-21, and miR-155 as potential biomarkers for HPV positive and negative cervical cancer

Authors:

Sunyoung Park¹† (sun-young@yonsei.ac.kr)
Kiyoon Eom¹† (richmond87@naver.com)
Jungho Kim¹ (jutosa70@hotmail.com)
Hyeeun Bang² (heybang@naver.com)
Hyeyoung Wang² (sapphire111@hanmail.net)
Sungwoo Ahn¹ (fast1945@hotmail.com)
Geehyuk Kim¹ (hyugiyoebi@naver.com)
Hyoungsoon Jang¹ (hushbee116@naver.com)
Sunghyun Kim³ (ksungh20@hotmail.com)
Dongsup Lee⁴ (eastern3547@naver.com)
Kwang Hwa Park⁵* (abba@yonsei.ac.kr)
Hyeyoung Lee¹* (hyelee@yonsei.ac.kr; angelsy88@gmail.com)

Version: 1 Date: 20 Sep 2016

Author’s response to reviews:

Thank you for the Editor's comments.

As you mention the revision of manuscript, the contents (consent and declaration) were revised or added as followed.

1 - Please indicate whether written informed consent was obtained from the human patients from whom the biopsy tissue was collected. If study approval or written human subject consent were not necessary, please state this in the manuscript and include a brief explanation.
In manuscript line 96, "Institutional Ethics Committee at Yonsei University Wonju College of Medicine approved the study protocol (approval no. YWMR-12-4-010) and all subjects provided written informed consent." was revised.

2 - Please note that all manuscripts must contain the following sections under the heading 'Declarations':

As you request the declaration, I added the information.

This is added in manuscript line 294-321.

Ethics approval and consent to participate

Institutional Ethics Committee at Yonsei University Wonju College of Medicine approved the study protocol (approval no. YWMR-12-4-010) and all subjects provided written informed consent.

Consent for publication

Not applicable.

Availability of data and material

All relevant materials are described in the manuscript. Additional data sets supporting the conclusions of this article are available at request from the corresponding author.

Competing interests

The authors declare that they have no competing interests.

Author’s contribution

SP, KE, and JK participated in the design of the study and performed the statistical analysis and manuscript. KE and HJ and HW carried out the data extraction. HY and KP and HB conceived of the study, and participated in its design and coordination and helped to draft the manuscripts. GK, DL, and SK helped to assemble the clinical data and performed the initial data interpretation together with HB and organize the experiment and sample collection. All authors read and approved the final manuscript.

Acknowledgements
This research was supported by Basic Science Research Program through the National Research Foundation of Korea (NRF) funded by the Ministry of Science, ICT and future Planning (2015R1A2A2A04004455).